Author's response to reviews

Title: Patient Presenting with Tentorial Schwannoma Unrelated to the Cranial Nerves

Authors:

Pietro Ivo D'Urso (pietrivo@me.com)
Michele Marino (marinomichele@libero.it)
Arturo Di Blasi (adiblasi@libero.it)
Carmine Franco Muccio (cfmuccio@libero.it)
Pompilio De Cillis (decil@hotmail.com)
Giuseppe Catapano (gcatapano@gmail.com)

Version: 4 Date: 1 November 2011

Author's response to reviews: see over
Dear Editor,

Thank you for your kind revision.

We are submitting the revised form of the manuscript **1774447265572161**.

We modified title, text and figures as requested.

Kind regards,

Pietro Ivo D’Urso

On behalf of all authors